Author's response to reviews

Title: Unbearable suffering in end-of-life cancer patients A cross-sectional study in primary care in the Netherlands

Authors:

Cees DM Ruijs (c.ruys3@kpnplanet.nl)
Ad JFM Kerkhof (ajfm.kerkhof@vu.nl)
Gerrit van der Wal (g.vanderwal@vumc.nl)
Bregje D Onwuteaka-Philipsen (b.philipsen@vumc.nl)

Version: 5 Date: 17 June 2012

Author's response to reviews: see over
Dear editor,

We are glad the article now can be accepted for publication and present version 4. The article has been checked and changes have been made in the formatting. I hope the tables are satisfactory; as a clinician the tricks of editing of tables are not my daily work. If further formatting of the tables is necessary I will need to ask for some assistance at the department.

The present study has been selected for poster presentation at the MASCC/ISOO International symposium on supportive care in cancer, New York City, June 28-30, 2012.

We hope you will find the concerns to be satisfactory addressed. If so, I hope for a formal reference soon, so I can add the reference to the poster.

The bill for the publication fee may be addressed (invoice) to:

Bregje Onwuteaka-Philipsen
VUmc
EMGO
Van der Boechorststraat 7
1081 BT Amsterdam
The Netherlands

Sincerely,
Also on behalf of Ad Kerkhof, Gerrit van der Wal and Bregje Onwuteaka-Philipsen,

Cees Ruijs, presenting author